\newcommand{\etalchar}[1]{$^{#1}$}
\begin{thebibliography}{{SPA}92b}

\bibitem[AS92]{Andr92}
K.~Andrews and D.~Sand.
\newblock Migrating a {CISC} computer family onto {RISC} via object code
  translation.
\newblock In {\em Proceedings {ASPLOS V}}, pages 213--222, October 1992.

\bibitem[ASU86a]{Aho86}
A.V. Aho, R.~Sethi, and J.D. Ullman.
\newblock {\em Compilers: Principles, Techniques, and Tools}.
\newblock Addison-Wesley, Reading, Massachusetts, 1986.

\bibitem[ASU86b]{Aho86a}
A.V. Aho, R.~Sethi, and J.D. Ullman.
\newblock {\em Compilers: Principles, Techniques, and Tools}.
\newblock Addison-Wesley Publishing Company, 1986.

\bibitem[BD88]{Beni88}
M.E. Benitez and J.W. Davidson.
\newblock A portable global optimizer and linker.
\newblock In {\em Proceedings of the Conference on Programming Languages,
  Design and Implementation}, pages 329--338. {ACM} Press, July 1988.

\bibitem[BD95]{Bail95}
M.W. Bailey and J.W. Davidson.
\newblock A formal model and specification language for procedure calling
  conventions.
\newblock In {\em {ACM S}ymposium on Principles of Programming Languages},
  pages 298--310, San Francisco, {CA}, January 1995.

\bibitem[BDB00]{Bala00}
V.~Bala, E.~Duesterwald, and S.~Banerjia.
\newblock Dynamo: A transparent dynamic optimization system.
\newblock In {\em Proceedings of the {ACM} {SIGPLAN}'00 Conference on
  Programming Language Design and Implementation}, pages 1--12, Vancouver,
  Canada, June 2000. {ACM P}ress.

\bibitem[Bed90]{Bedi90}
R.~Bedichek.
\newblock Some efficient architecture simulation techniques.
\newblock In {\em Proceedings Winter {USENIX} Conference}, pages 53--63, 1990.

\bibitem[Ber85]{Bern85}
R.L. Bernstein.
\newblock Producing good code for the case statement.
\newblock {\em Software--Practice and {E}xperience}, 15(10):1021--1024, 1985.

\bibitem[BKMM87]{Berg87}
A.B. Bergh, K.~Keilman, D.J. Magenheimer, and J.A. Miller.
\newblock {HP3000} emulation on {HP} precision architecture computers.
\newblock {\em Hewlett-Packard Journal}, pages 87 -- 89, December 1987.

\bibitem[CE99]{Cifu99c}
C.~Cifuentes and M.~Van Emmerik.
\newblock Recovery of jump table case statements from binary code.
\newblock In {\em Proceedings of the International Workshop on Program
  Comprehension}, pages 192--199, Pittsburgh, {USA}, May 1999. {IEEE CS P}ress.

\bibitem[CF97]{Cifu97d}
C.~Cifuentes and A.~Fraboulet.
\newblock Intraprocedural static slicing of binary executables.
\newblock In M.J. Harrold and G.~Visaggio, editors, {\em Proceedings of the
  International Conference on Software Maintenance}, pages 188--195, Bari,
  Italy, 1--3 October 1997. {IEEE-CS P}ress.

\bibitem[CG95]{Cifu95}
C.~Cifuentes and K.J. Gough.
\newblock Decompilation of binary programs.
\newblock {\em Software -- Practice and Experience}, 25(7):811--829, July 1995.

\bibitem[Cha91]{Cham91}
S.~Chamberlain.
\newblock {\em libbfd -- The Binary File Descriptor Library}, first edition --
  {BFD} version 3.0 edition, April 1991.

\bibitem[Chi95]{Chia95}
L.K. Chia.
\newblock Binary emulation.
\newblock Master's thesis, Purdue University, Department of Electrical
  Engineering, May 1995.

\bibitem[Cif96]{Cifu96}
C.~Cifuentes.
\newblock Interprocedural dataflow decompilation.
\newblock {\em Journal of Programming Languages}, 4(2):77--99, June 1996.

\bibitem[CK94]{Cmel94}
B.~Cmelik and D.~Keppel.
\newblock Shade: A fast instruction-set simulator for execution profiling.
\newblock In {\em Proceedings {ACM SIGMETRICS} Conference on Measurement and
  modeling of Computer Systems}, 1994.

\bibitem[CM96]{Cifu96c}
C.~Cifuentes and V.~Malhotra.
\newblock Binary translation: Static, dynamic, retargetable?
\newblock In {\em Proceedings of the International Conference on Software
  Maintenance}, pages 340--349, Monterey, CA, 4-8 November 1996. IEEE Computer
  Society Press.

\bibitem[Cog95]{Cogs95b}
B.H. Cogswell.
\newblock {\em Timing Insensitive Binary-to-Binary Translation}.
\newblock {PhD} dissertation, Carnegie Mellon University, Department of
  Electrical and Computer Engineering, April 1995.

\bibitem[Cor94]{Apple94}
Apple Corporation.
\newblock Macintosh application environment.
\newblock \url{http://www.mae.apple.com}, 1994.

\bibitem[CS95]{Cogs95}
B.H. Cogswell and Z.~Segall.
\newblock Timing insensitive binary-to-binary migration across multiprocessor
  architectures.
\newblock In {\em Proceedings of the Third Workshop on Parallel and Distributed
  Real-Time Systems}, pages 193--194, Santa Barbara, CA, USA, April 1995.
  {IEEE} Computer Society Press.

\bibitem[CS98]{Cifu98c}
C.~Cifuentes and S.~Sendall.
\newblock Specifying the semantics of machine instructions.
\newblock In {\em Proceedings of the International Workshop on Program
  Comprehension}, pages 126--133, Ischia, Italy, 24--26 June 1998. {IEEE CS
  P}ress.

\bibitem[CS99]{Cifu99g}
C.~Cifuentes and D.~Simon.
\newblock Procedural abstraction recovery from binary code.
\newblock Technical Report 448, The University of Queensland, Department of
  Computer Science and Electrical Engineering, September 1999.

\bibitem[CSF98a]{Cifu98h}
C~Cifuentes, D~Simon, and A~Fraboulet.
\newblock Assembly to high-level language translation.
\newblock In {\em Proceedings of the International Conference on Software
  Maintenance}, pages 228--237, Washington {DC}, {USA}, 18--20 November 1998.
  {IEEE CS P}ress.

\bibitem[CSF98b]{cifuentes:assembly}
Cristina Cifuentes, Doug Simon, and Antoine Fraboulet.
\newblock Assembly to high-level language translation.
\newblock In {\em Proceedings of the International Conference on Software
  Maintenance}, pages 228--237. IEEE-CS Press, November 1998.

\bibitem[Dig94]{Dec94}
Digital.
\newblock {OM} and {ATOM}.
\newblock \url{http://www.research.digital.com/wrl/projects/om/om.html}, 1994.
\newblock Digital Western Research Labs.

\bibitem[Dig95]{Dec95}
Digital.
\newblock Freeport express.
\newblock \url{http://www.novalink.com/freeport-express}, 1995.

\bibitem[DMP96]{danvy:eta}
Olivier Danvy, Karoline Malmkj{\ae}r, and Jens Palsberg.
\newblock Eta-expansion does {The Trick}.
\newblock {\em ACM Transactions on Programming Languages and Systems},
  18(6):730--751, November 1996.

\bibitem[DR97]{Duke97}
R.~Duke and G.~Rose.
\newblock Formal object-oriented specification and design using {O}bject-{Z}.
\newblock Book to be published, 1997.

\bibitem[Dun88]{Dunc88b}
R.~Duncan.
\newblock {\em The MSDOS Encyclopedia}, chapter~4, pages 107--147.
\newblock Microsoft Press, 1988.

\bibitem[EA96]{Ebci96}
K.~Ebcioglu and E.~Altman.
\newblock {DAISY}: Dynamic compilation for 100\% architectural compatibility.
\newblock Technical Report {RC 20538}, {IBM}, {IBM T.J. W}atson Research
  Center, Yorktown Heights, New York, August 1996.

\bibitem[ES95]{Eust95}
A.~Eustace and A.~Srivastava.
\newblock {ATOM} a flexible interface for building high performance program
  analysis tools.
\newblock In {\em Proceedings {USENIX} Technical Conference}, pages 303--314,
  January 1995.
\newblock Also as Digital Western Research Laboratory Technical Note {TN-44},
  July 1994.

\bibitem[ESL88]{Emme88}
H.~Emmelmann, F.~Schroer, and R.~Landwehr.
\newblock {BEG - A} generator for efficient back ends, 1988.

\bibitem[Fer95]{Fern95}
M.~Fern\'{a}ndez.
\newblock Simple and effective link-time optimization of modula-3 programs.
\newblock In {\em SIGPLAN Conference on Programming Languages, Design and
  Implementation}, pages 103--115, June 1995.

\bibitem[FH91]{Fras91}
C.W. Fraser and D.R. Hanson.
\newblock A retargetable compiler for {ANSI C}.
\newblock {\em {SIGPLAN} Notices}, 26(10):29--43, October 1991.

\bibitem[FH95]{Fras95}
C.W. Fraser and D.R. Hanson.
\newblock {\em A Retargetable {C} Compiler: Design and Implementation}.
\newblock The Benjamin/Cummings Publishing Company, Inc., Redwood City, CA,
  1995.

\bibitem[FJ88]{Fisc88b}
C.N. Fischer and R.J.~LeBlanc Jr.
\newblock {\em Crafting a Compiler}.
\newblock Benjamin Cummings, 2727 Sand Hill Road, Menlo Park, California 94025,
  1988.

\bibitem[GAS{\etalchar{+}}00]{Gsch00}
M.~Gschwind, E.~Altman, S.~Sathaye, P.~Ledak, and D.~Appenzeller.
\newblock Dynamic and transparent binary translation.
\newblock {\em Computer}, 33(3):54--59, March 2000.

\bibitem[GP00]{Gepp00}
L.~Geppert and T.S. Perry.
\newblock Transmeta's magic show.
\newblock {\em {IEEE S}pectrum}, pages 26--33, May 2000.

\bibitem[HH97]{Hook97}
R.J. Hookway and M.A. Herdeg.
\newblock Digital {FX}!32: Combining emulation and binary translation.
\newblock {\em Digital Technical Journal}, 9(1):3--12, 1997.

\bibitem[HM79]{Hors79}
R.N. Horspool and N.~Marovac.
\newblock An approach to the problem of detranslation of computer programs.
\newblock {\em The Computer Journal}, 23(3):223--229, 1979.

\bibitem[HM82]{Henn82}
J.L. Hennessy and N.~Mendelsohn.
\newblock Compilation of the {P}ascal case statement.
\newblock {\em Software--Practice and {E}xperience}, 12:879--882, 1982.

\bibitem[Hol95]{Holz95}
U.~Holzle.
\newblock {\em Adaptive Optimization for {SELF}: Reconciling High Performance
  with Exploratory Programming}.
\newblock {PhD} dissertation, Stanford University, Department of Computer
  Science, March 1995.

\bibitem[HTB{\etalchar{+}}98]{Huan98}
H.~Huang, W.~Tsai, S.~Bhattacharya, X.~Chen, and Y.~Wang.
\newblock Business rule extraction techniques for {COBOL} programs.
\newblock {\em Journal of Software Maintenance: Research and Practice},
  10(1):3--35, January 1998.

\bibitem[ida97]{Ida97}
Ida {PRO} disassembler, {D}ata {R}escue.
\newblock \url{http://www.datarescue.com/ida.htm}, 1997.

\bibitem[{IEE}93]{Iee93}
{IEEE}.
\newblock {\em {IEEE} Standard 1219-1993 for Software Maintenance}.
\newblock {IEEE}, 1993.

\bibitem[Joh90]{John90}
S.C. Johnson.
\newblock Postloading for fun and profit.
\newblock In {\em Proceedings Winter {USENIX} Conference}, pages 325--330,
  1990.

\bibitem[JR99]{Peyt99}
S.~Peyton Jones and N.~Ramsey.
\newblock Machine-independent support for garbage collection, debugging,
  exception handling, and concurrency (draft), 1999.

\bibitem[Kan88]{kane:mips}
Gerry Kane.
\newblock {\em {MIPS} {RISC} Architecture}.
\newblock Prentice Hall, Englewood Cliffs, NJ, 1988.

\bibitem[LB94]{Laru94}
J.R. Larus and T.~Ball.
\newblock Rewriting executable files to measure program behavior.
\newblock {\em Software -- Practice and Experience}, 24(2):197--218, February
  1994.

\bibitem[LS95]{Laru95}
J.R. Larus and E.~Schnarr.
\newblock {EEL}: Machine-independent executable editing.
\newblock In {\em SIGPLAN Conference on Programming Languages, Design and
  Implementation}, pages 291--300, June 1995.

\bibitem[May87]{May87}
C.~May.
\newblock {MIMIC}: A fast {S}ystem/370 simulator.
\newblock In {\em Proceedings {SIGPLAN}'87 Symposium on Interpreters and
  Interpretive Techniques}, pages 1--13, June 1987.

\bibitem[Mey97]{Meye97}
J.~Meyer.
\newblock Jasmin.
\newblock \url{http://cat.nyu.edu/meyer/jasmin/}, 1997.

\bibitem[Mic]{Micr96}
Microsoft.
\newblock Microsoft windows software development kit {(SDK)} for windows.
\newblock Article ID: Q65260.

\bibitem[Mil90]{milner:operational}
Robin Milner.
\newblock Operational and algebraic semantics of concurrent processes.
\newblock In J.~van Leewen, editor, {\em Handbook of Theoretical Computer
  Science}, volume B: Formal Models and Semantics, chapter~19, pages
  1201--1242. The MIT Press, New York, N.Y., 1990.

\bibitem[Pro95]{Proe95}
T.A. Proebsting.
\newblock Optimizing and {ANSI C} interpreter with superoperators.
\newblock In {\em Proceedings of the 22nd Symposium on Principles of
  Programming Languages}, pages 322--332. {ACM} Press, January 1995.

\bibitem[RC98]{Cifu98i}
N.~Ramsey and C.~Cifuentes.
\newblock A transformational approach to binary translation of delayed
  branches.
\newblock Technical Report 440, The University of Queensland, Department of
  Computer Science and Electrical Engineering, December 1998.

\bibitem[RD98]{ramsey:embedded}
Norman Ramsey and Jack~W. Davidson.
\newblock Machine descriptions to build tools for embedded systems.
\newblock In {\em ACM SIGPLAN Workshop on Languages, Compilers, and Tools for
  Embedded Systems (LCTES'98)}, volume 1474 of {\em LNCS}, pages 172--188.
  Springer Verlag, June 1998.

\bibitem[RDGY00]{Reev00}
C.~Reeve, D.~Deaver, R.~Gorton, and B.~Yadavalli.
\newblock Wiggins/redstone (wr): A dynamic optimization and specialization
  tool.
\newblock Unpublished manuscript, 2000.

\bibitem[RF94a]{Rams94}
N.~Ramsey and M.~Fern\'{a}ndez.
\newblock {\em New {J}ersey Machine-Code Toolkit Reference Manual}, version
  0.1a edition, November 1994.
\newblock URL ftp://ftp.cs.princeton.edu/reports/1994/471.ps.Z.

\bibitem[RF94b]{Rams94b}
N.~Ramsey and M.F. Fern\'{a}ndez.
\newblock The {N}ew {J}ersey machine-code toolkit.
\newblock Technical Report~64, Princeton University, Department of Computer
  Science, Princeton, NJ 08544, 1994.

\bibitem[RF95]{Rams95}
N.~Ramsey and M.~Fern\'{a}ndez.
\newblock The {N}ew {J}ersey machine-code toolkit.
\newblock In {\em Proceedings of the 1995 {USENIX} Technical Conference}, pages
  289--302, January 1995.

\bibitem[RF97a]{Rams97}
N.~Ramsey and M.~Fern\'{a}ndez.
\newblock Specifying representations of machine instructions.
\newblock {\em {ACM} Transactions of Programming Languages and Systems},
  19(3):492--524, 1997.

\bibitem[RF97b]{ramsey:specifying}
Norman Ramsey and Mary~F. {Fer\-n\'an\-dez}.
\newblock Specifying representations of machine instructions.
\newblock {\em ACM Transactions on Programming Languages and Systems},
  19(3):492--524, May 1997.

\bibitem[RH92]{Rams92}
N.~Ramsey and D.R. Hanson.
\newblock A retargetable debugger.
\newblock In {\em Proceedings Programming Languages, Design and
  Implementation}, pages 22--31, June 1992.

\bibitem[Sal81]{Sale81}
A.~Sale.
\newblock The implementation of case statements in {P}ascal.
\newblock {\em Software--Practice and {E}xperience}, 11:929--942, 1981.

\bibitem[SCK{\etalchar{+}}92]{Site92}
R.L. Sites, A.~Chernoff, M.B. Kirk, M.P. Marks, and S.G. Robinson.
\newblock Binary translation.
\newblock {\em Digital Technical Journal}, 4(4):1--16, 1992.

\bibitem[SCK{\etalchar{+}}93]{Site93}
R.L. Sites, A.~Chernoff, M.B. Kirk, M.P. Marks, and S.G. Robinson.
\newblock Binary translation.
\newblock {\em Communications of the ACM}, 36(2):69--81, February 1993.

\bibitem[SE93]{Silb93}
G.M. Silberman and K.~Ebcioglu.
\newblock An architectural framework for supporting heterogeneous
  instruction-set architectures.
\newblock {\em Computer}, pages 39--56, June 1993.

\bibitem[Sen97]{Send97}
S.~Sendall.
\newblock Semantics of machine instructions.
\newblock Honours thesis, The University of Queensland, Department of Computer
  Science and Electrical Engineering, 1997.

\bibitem[Sha96]{shankar:mechanizing}
Natarajan Shankar.
\newblock Steps towards mechanizing program transformations using {PVS}.
\newblock {\em Science of Computer Programming}, 26(1--3):33--57, May 1996.

\bibitem[spa92a]{Spar92}
Sparc International, Menlo Park, California.
\newblock {\em The {SPARC} Architecture Manual -- Version 8}, 1992.

\bibitem[{SPA}92b]{sparc:architecture}
{SPARC International}, Englewood Cliffs, NJ.
\newblock {\em The {SPARC} Architecture Manual, Version~8}, 1992.

\bibitem[{SPA}92c]{Sparc8}
{SPARC International Inc.}
\newblock {\em The {SPARC} Architecture Manual -- Version 8}.
\newblock Prentice Hall, Englewood Cliffs, New Jersey 07632, 1992.

\bibitem[Sta93]{Stal93}
R.M. Stallman.
\newblock {\em Using and Porting {GNU CC}}, version 2.5 edition, October 1993.

\bibitem[Sun94a]{Sun94m}
SunSoft.
\newblock {\em Linker and Libraries Guide -- Solaris 2.4}.
\newblock Sun Microsystems Inc., Mountain View, California, August 1994.

\bibitem[Sun94b]{Sun94}
SunSoft.
\newblock Wabi.
\newblock \url{http://www.sun.com/sunsoft/Products/PC-Integration-products},
  1994.

\bibitem[SW93a]{Sriv93}
A.~Srivastava and D.W. Wall.
\newblock A practical system for intermodule code optimization at link-time.
\newblock {\em Journal of Programming Languages}, 1(1):1--18, March 1993.

\bibitem[SW93b]{srivastava:practical}
Amitabh Srivastava and David~W. Wall.
\newblock A practical system for intermodule code optimization.
\newblock {\em Journal of Programming Languages}, 1:1--18, March 1993.
\newblock Also available as WRL Research Report 92/6, December 1992.

\bibitem[T94]{Att94}
AT~T.
\newblock Flashport.
\newblock \url{http://www.att.com/FlashPort}, 1994.
\newblock AT\&T Bell Labs.

\bibitem[TBS99]{Taiv99}
A.~Taivalsaari, B.~Bush, and D.~Simon.
\newblock The {S}potless system: Implementing a {J}ava$^{TM}$ system for the
  {P}alm connected organizer.
\newblock Technical Report {SMLI TR-99-73}, Sun Microsystems Laboratories, 901
  San Antonio Rd, Palo Alto, CA, February 1999.

\bibitem[The95]{Then95}
T.W. Then.
\newblock Binary translation with optimizations.
\newblock Master's thesis, Purdue University, Department of Electrical
  Engineering, December 1995.

\bibitem[Tho96]{Thom96}
T.~Thompson.
\newblock An {A}lpha in {PC} clothing.
\newblock {\em Byte}, pages 195--196, February 1996.

\bibitem[Tra97]{Trac97}
TracePoint.
\newblock {HiProf} hierarchical profiler.
\newblock \url{http://www.tracepoint.com/noframes/hiprof/products/hiprof},
  1997.

\bibitem[Ts'00]{Tso00}
Theodore Ts'o.
\newblock The prc format.
\newblock http://web.mit.edu/tytso/www/pilot/prc-format.html, 2000.

\bibitem[UC97]{Cifu97f}
D.~Ung and C.~Cifuentes.
\newblock {SRL} - a simple retargetable loader.
\newblock In {\em Proceedings of the Australian Software Engineering
  Conference}, pages 60--69, Sydney, Australia, 28 September - 2 October 1997.
  {IEEE CS P}ress.

\bibitem[uni90a]{Unix90}
{\em Unix {S}ystem {V}: Application Binary Interface}.
\newblock Prentice Hall, Englewood Cliffs, New Jersey, 1990.

\bibitem[uni90b]{Unix90b}
{\em Unix {S}ystem {V}: Application Binary Interface, {SPARC} Processor
  Supplement}.
\newblock Prentice Hall, Englewood Cliffs, New Jersey, 1990.

\bibitem[uni90c]{Unix90c}
{\em Unix {S}ystem {V}: Application Binary Interface, {Intel} Processor
  Supplement}.
\newblock Prentice Hall, Englewood Cliffs, New Jersey, 1990.

\bibitem[vpo98]{Vpo98}
Machine-independent optimization.
\newblock \url{http://www.cs.virginia.edu/zephyr/vpo}, 1998.

\bibitem[Wal92]{wall:systems}
David~W. Wall.
\newblock Systems for late code modification.
\newblock In Robert Giegerich and Susan~L. Graham, editors, {\em Code
  Generation - Concepts, Tools, Techniques}, pages 275--293. Springer-Verlag,
  1992.

\bibitem[WH66]{Wirt66}
N.~Wirth and C.A.R. Hoare.
\newblock A contribution to the development of {ALGOL}.
\newblock {\em Communications of the {ACM}}, 9(6):413--432, 1966.

\bibitem[win96]{Wine96}
The wine project.
\newblock \url{http://www.winehq.com}, 1996.

\bibitem[WLG94]{Wahb94}
R.~Wahbe, S.~Lucco, and S.L. Graham.
\newblock Adaptable binary programs.
\newblock Technical Report {CMU-CS-94-137}, Carnegie Mellon University, School
  of Computer Science, Pittsburgh, PA 15213, April 1994.

\bibitem[Wra74]{Wran74}
C.~Wrandle.
\newblock Notes on the \texttt{case} statement.
\newblock {\em Software--Practice and {E}xperience}, 4:289--298, 1974.

\bibitem[ZT00]{Zhen00}
C.~Zheng and C.~Thompson.
\newblock {PA-RISC} to {IA-64}: Transparent execution, no recompilation.
\newblock {\em Computer}, 33(3):47--52, March 2000.

\end{thebibliography}


\chapter{The BinaryFile and ArchiveFile classes}
\label{ch-bff}

{\small
\begin{flushright}
Design: Cristina, Mike; Documentation: Cristina, Mike; Implementation: Mike
[c.1997]
\end{flushright}
}

The term "loader" is generally used to describe a system program
used by an operating system (OS) to load a
binary executable file onto memory to execute it. 
We have a class called "BinaryFile" that can be used by application
programs to load other binary files for purposes other than directly
executing them. (The class was formerly called "Loader", but this
contrasts with the use above).
In other words, BinaryFile is a decoder of binary-file formats; it 
reads a binary executable file and stores its representation in memory,
providing functions to access the different parts of the binary
file; such as code and data.  
It provides extra functionality in the presence of 
dynamically linked-in procedures; binary-file formats such as
ELF and PE (Portable Executable) support them.  In this case,
the BinaryFile interface provides a way of determining if a 
procedure address is an address for a dynamically linked-in 
procedure, and if so, it allows access to its code and data.
In this regard, the BinaryFile class is more of a dynamic linker/loader.

Binary files vary widely in internal organisation and structure,
nevertheless, they provide similar kinds of information 
in order to run the program.  
The main components of a binary file are its code and its data;
everything else is a representational structure to access this
information.
By means of the BinaryFile and ArchiveFile classes, we attempt to 
provide a uniform interface for the loading and usage of the 
information stored in binary files.

Some binary file of interest are collected in library or archive
files. The members of the archive are usually object (.o) files;
there is usually a symbol table associated with the archive so that
the member containing the symbol can readily be found. Archive
files are obviously used rather differently than other binary files
(executable and object files), despite attempts to unify them (e.g.
in the elflib library). Therefore, functions for using archive
files are separated into their own class, called ArchiveFile.
(In the previous form, class Loader had a function GetNextMember
to move to the next member of an archive). When a member of the
archive is selected (by index, or procedure name, or file name),
a reference to an instance of a  BinaryFile class is returned, and all the
BinaryFile functions can be called (except for Load; the BinaryFile
object comes "preloaded").


\section{Related Work}
We briefly describe the two main pieces of related work in this
area.

\subsection{GNU's Binary File Descriptor Library}
GNU's Binary-File Descriptor (BFD) Library~\cite{Cham91} is a package
containing common routines that applications can use regardless of 
their underlying binary-file format.
The BFD library divides each specified BFF into the front-end and the
back-end.
The front-end interfaces between the user and the BFD, while
the back-end provides a set of calls which the BFD front-end can use to
decode and manage the object file.
To support a new BFF, the programmer needs to create a new BFD back-end
and add it to the library.
 
BFD has its own binary representation for internal processing known as the
canonical object file format.
When an binary file is opened, the front-end BFD routines automatically 
determine the format of the input file.  A descriptor is built in memory 
with information about which routines are to be used to access
elements of the binary file's data structure.  When the program wants
information about the binary files, the BFD reads from different sections of
the file and processes them.  Each BFD back-end will have routines to convert
section representations of the binary file to BFD's internal canonical
object-file format.
 
The BFD library is provided to the user as a library.  This library
is fairly large; the number of functions offered in the front-end 
are exceptionally many.
The BFD front-end was designed in mind to allow programmers to
be able to retrieve all types of information about \emph{any} BFF; at
least the existing ones at the time.
Due to its generality and bulkiness, it is difficult to use without
spending a big overhead on learning how to use it.
Perhaps because it is too general, it often contain more information 
than is needed for particular system applications.
 

\subsection{SRL - A Simple Retargetable Loader}
SRL, a simple retargetable loader, is a first attempt at developing a 
retargetable loader framework by means of a simple BFF grammar~\cite{Cifu97f}.  
Three different environments, (x86,DOS,EXE), (x86,Windows,NE) and 
(Sparc,Solaris,ELF), were used as the basis for the development and 
testing of SRL.  The three environments gave a good coverage of different 
BFFs currently in use by OSs for RISC and CISC machines. 

The BFF grammar provides support for describing sections of
the binary file, and to name different fields from each section.
Further, it provides support for structures which have been
stored as a sequence of records of a given type, e.g. the number of
elements in the segment table on the NE (New Executable) format,
by means of an array construct -- this construct clearly aids in
the specification of BFFs.

The EBNF for this grammar's syntax is provided in Figure~\ref{fig-bffg}.  
In the grammar, {\it non-terminals} appear in italics, terminals appear in
normal fontface, ``literal strings'' appear with double quotes, and
\verb!examples! appear in courier.
The start symbol for this grammar is {\it BFFspec}.
Hence, the body for any BFF specification is of the form: \\
{\small
{\it spec} $=>$ {\it format-def defin \{defin\} load-info}
}

\centerfigbegin
{\small
\begin{tabular}{lll}
 {\it BFFspec} & $=>$ & {\it \{spec\}}. \\
 {\it spec} & $=>$ & {\it format-def defin \{defin\} load-info} \\
 {\it format-def} & $=>$ & ``DEFINITION'' ``FORMAT'' \\
    & & {\it ident \{ident\}} ``END'' ``FORMAT'' \\
 {\it defin} & $=>$ & ``DEFINITION'' {\it ident} \\
    & & ``ADDRESS'' {\it expression scope-def} \\
    & & ``END'' {\it ident}. \\
 {\it load-info} & $=>$ & ``FILEHEADER'' {\it ident} \\
     & & ``IMAGESIZE'' {\it expression} \\
     & & ``IMAGEADDRESS'' {\it expression} \\
 {\it scope-def} & $=>$ & {\it ident type-exp \{ident type-exp\}} \\
 {\it type-exp} & $=>$ & ``SIZE'' {\it expression} $|$ \\
    & & ``ARRAY'' {\it expression scope-def} \\
    & & ``END'' {\it ident} \\
 {\it expression} & $=>$ & ``('' {\it ident operator expression} ``)'' \\
     & & $|$ {\it ident operator expression} $|$ $\epsilon$ \\
 {\it operator} & $=>$ & ``+'' $|$ ``-'' $|$ ``*'' $|$ ``/'' $|$ ``\^'' $|$ ``\%'' \\
 {\it ident} & $=>$ & ``a''..``z'' $|$ ``A''..``Z'' \{``a''..``z'' $|$ \\
    & & ``A''..``Z'' $|$ ``\_''\} \\
\end{tabular}
}
\centerfigend{fig-bffg}{Binary-File Format Grammar}

SRL, the tool that implements the BFF grammar, was an attempt
to demonstrate the benefit of using a retargetable loader to 
build a machine-code manipulation tool.  SRL was limited in 
a way by its simple grammar which contained a small number of
constructs.  Nevertheless, the BFF grammar was suitable for
specifying most of the sections of the ELF, EXE and NE formats.
SRL generates a C code in the form of a header file (.h) and 
an implementation (.c) file from each parsed file written in
the BFF language.  The header file contains the data structures
needed for the storing of information for a particular
binary file format, and the implementation file provides 
functions for the loading of a file using the structures 
defined in the header file. 


\subsection{Our Approach}
Our approach is different to the two previous ones, as it is more
specific and less retargetable.  We provide an API (via the
BinaryFile class) that 
users must adhere to, but we do not generate code for it automatically
from specifications, nor do we provide for a complete interface
suitable for a large number of binary file formats.  In contrast,
we provide an object oriented abstraction which provides the base
functionality of a loader (regardless of binary file format) in 
an abstract class, and loaders for specific binary file formats 
(e.g. EXE or ELF) inherit from this abstract class and provide new
functionality specific to their format. Similarly, there is an
ArchiveFile class that defines functions for using archive files,
and there are classes derived from this class for the various types
of archive file.


\section{Binary-file formats}
\label{sec-bff}
We briefly describe the abstract format of a binary file.  
Users not familiar with the internal representation of 
binary files who want more detailed information may refer
to the following literature: \cite{Dunc88b,Sun94m,Micr96} and
the web site \url{http://www.wotsit.demon.co.uk} which has a 
compendium of binary file formats.

The general structure of a binary file format (BFF) can be seen to 
be made up by the following abstraction:
\begin{itemize}
\item A header containing general information about the program
and information needed to access various parts of the file.
\item A number of sections holding code and data (raw data).
\item A relocation table containing offsets of relocatable addresses.
\item A symbol table containing information about symbols of the 
	program.
\end{itemize}
Each of these \emph{parts} is given a name respectively: \emph{header},
\emph{sections}, \emph{relocation table} and \emph{symbol table}.
Further, some binary files such as ELF allow for more than one linked
segment to be stored in the one physical file or archive; we refer
to these segments as \emph{members} of the archive.

Most BFFs can be mapped to the general model in Figure~\ref{fig-bffoa};
however, parts are not necessarily stored in that order.
Information regarding the location of sections, symbol tables, etc is
usually identified within the file header.
Nevertheless, some BFFs do not distinguish between these structures; in
the DOS EXE format, the file header contains information about the relocation
table, but there is no information about where the symbol table is stored
(if any), and where data is; there is only one section that embodies all
code, data and symbol table information.
In all cases though, the program's
header will contain enough information to determine the entry point (i.e.
the start of the program's code) in the file.

% Original file in Visio format, on Blimey (in the Drawings directory)
\centerfigbegin
\resizebox{!}{5cm}
{\includegraphics{figures/bffoa.eps}}
\centerfigend{fig-bffoa}{BFF and archive file abstraction}

The current development domain for our tools is based on the Solaris 
ELF~\cite{Sun94m} format, the DOS EXE format~\cite{Dunc88b,Micr96}, 
the Windows (16-bit) NE~\cite{Dunc88b,Micr96} format,
and the Palm OS .prc format~\cite{Tso00}.
Archive files are based on the Unix ar(4) file format.
These formats vary in their degree of complexity and information
stored: the DOS EXE is very simple and limited in structure,
whereas the Solaris ELF format is the most complex, while the Windows
NE is somewhere in between.
The amount of information stored for a simple ``Hello world'' program
varies from format to format.  The DOS EXE format contains a file
header, a relocation table and a single image for both code and data.
The Windows NE version contains most DOS EXE's information plus
additional details such as the resource table, entry table, etc.
The ELF format contains even more information; sections within the 
object file hold information used in dynamic linking: code, data, 
relocation tables, symbol tables, dynamic linking information, etc.
The size in bytes of the binaries for (x86,DOS,EXE), (x86,Windows,NE)
and (Sparc,Solaris,ELF) are 6432, 16384 and 5280 respectively.
It can clearly be seen that although the latter two files are dynamically
linked, their sizes are not necessarily smaller than the static (first)
case.
This is due to the small nature of the example program and the
inclusion of the DOS EXE header information within the NE format.



\section{The BinaryFile Object Hierarchy}
BinaryFile and ArchiveFile are abstract classes; that is, they cannot be
instantiated
directly. The user actually uses classes such as ElfBinaryFile or ExeBinaryFile,
which are derived from the abstract BinaryFile class (see 
Figure~\ref{fig-loadHier}).

% xfig figures exported as eps, portrait, 100% size, displayed at 3.2cm.
\centerfigbegin
\resizebox{!}{3.2cm}
{\includegraphics{figures/load_hierarchy.eps}}
\centerfigend{fig-loadHier}{BinaryFile Class Hierarchy}

The user will call functions like \texttt{Load} that are implemented 
differently in the various derived classes. Some functions like
\texttt{SymbolByAddress} are implemented trivially by the BinaryFile class 
(e.g. to always return 0), and this behaviour is overridden by derived 
classes which do the real work.  Even functions like \texttt{GetNumSyms} 
that are implemented by the BinaryFile class will return information that 
will be dependent on the instantiated class.



\section{Interface Functions to Construct and Use a BinaryFile}
The BinaryFile class and each of its parts (described in the abstract 
format description of a binary file in section~\ref{sec-bff}), are 
explained here in terms of interface functions provided for accessing 
information in those parts. 


\subsection{Construction and Loading}
The BinaryFile class provides constructor and destructor 
functions, plus a function to load the binary file.
The loader is an abstract class and hence it does not export 
any types.

\begin{description}
\item [BinaryFile: BOOL $\rightarrow$ BinaryFile].
	This is the constructor. The single BOOL argument is used to
	indicate whether the object to be constructed is to be an
	Archive member; it defaults to false. Users should not use
	this parameter, but should use the ArchiveFile class
	instead (see section~\ref{sec-archive}). The BinaryFile object
	is created without a file loaded; use the Load() function below
	to actually load a file.
	
\item [Load: (BinaryFile x STRING) $\rightarrow$ BOOL].
	Load the given file and store its
	information in a BinaryFile object. It returns whether the
	function was successful or not.
	If there is a problem loading the file, e.g. the file name does
	not exist or there is not enough memory, a message is printed
	on standard error.

\item [UnLoad: BinaryFile $\rightarrow$ nil].
	Deallocates the BinaryFile object. All information about the file
	(e.g. SectionInfo, pointers to section data) are invalid after
	calling this function.

\end{description}


\subsection{Sections}
For each section in the binary file, the following information is 
collected in a SECTIONINFO type:
\begin{itemize}
\item Section name: the name of the section, if any.  Formats like
	ELF support names for each section, such as \texttt{.text} and
    \texttt{.bss}; formats such as EXE do not provide such information
	and hence we follow the convention of naming them \texttt{\.header}
	and \texttt{.text} for the header of the file and the relocatable
	image respectively.

\item Native address: address where the section is logically loaded.
	For example, all EXE files are loaded at address 0x100000,
	representing 10:0000.  ELF files are loaded at the address
	specified in the elf file, typically 0x10000 for code and 0x20000
	for data.  The native address is typically used by diassemblers
	to give a useful address in listing. 

\item Host address: the host address is where the section is actually
	located in virtual memory, i.e. this is a pointer to the first
	byte in the section.

\item Section size: size of the section in bytes.

\item Number of entries: if the section is the equivalent to an 
	array of entries, the size of each entry is given.

\item bCode: This flag is set if the section contains executable
	instructions.

\item bData: This flag is set if the section contains data. It is possible
	for both bCode and bData to be set at the same time.

\item bBss: This flag is set if the section is not initialised, i.e. it
	is for reserving space only.

\item bReadOnly: This flag is set if the section is read only, i.e. it
	cannot be written to.

\end{itemize}

For example, with the ElfBinaryFile class, the section whose name is 
\texttt{\.rodata} would have the flags bData and bReadOnly set.
A disassembler can find all the sections worth disassembling by 
searching through all the sections for those with the flag bCode set.

Sections provide the following functions to determine the number
of sections available in the file and access its information:
\begin{description}
\item [GetNumSections: BinaryFile $\rightarrow$ int].
	Given a BinaryFile object, returns the number of sections that it 
	contains.  

\item [GetSectionInfo: (BinaryFile x int) $\rightarrow$ SECTIONINFO].
	Given an BinaryFile object and a section index, returns the information
	for that section.  Note that sections are indexed from 1.
\end{description}


\subsection{Symbol Table}
Symbol tables are often used for dynamic linking purposes, where
the names of dynamically linked-in routines are stored.  However,
sometimes program symbols are also stored in a symbol table,
whether in the same or a separate table to that of dynamic linking.

The symbol table offered by the BinaryFile class is very simple; 
there are three types of information stored for each entry:
\begin{itemize}
\item Symbol name: the symbol's name in the table.  Type: STRING.

\item Symbol address: the symbol's native address.  Type: ADDRESS.

\item Symbol size: the size in bytes associated with the symbol.
	Type: int.
\end{itemize}

The symbol table offers three access functions to its contents:
\begin{description}
\item [SymbolByAddress: (BinaryFile x ADDRESS) $\rightarrow$ STRING].
	Given a native address, returns the symbolic name associated with the 
	given native address.
	This function is useful when getting names for procedures at
	call locations.

\item [GetAddressByName: (BinaryFile x STRING) $\rightarrow$ ADDRESS].
	Given the name of a symbol, returns the native address associated
	with it.	
	If the symbol is inexistent in the symbol table, an address of 
	0 is returned.

\item [GetSizeByName: (BinaryFile x STRING) $\rightarrow$ int].
	Given the name of a symbol, returns the size in bytes of the 
	symbol, or 0 if inexistent.
\end{description}

Not all BinaryFiles provide the ByName() functions, because in some
cases the binary file format does not support it. If the functions are
not supported, they will always return 0.

Some BinaryFile classes (e.g. ExeBinaryFile) do not implement a symbol table.


\subsection{Relocation Table}
Relocation tables provide information on address that need to be
relocated prior to execution of the program.  
The advantage of knowing which addresses are relocatable is
that when decoding a machine instruction that takes an immediate
operand, we can check if this operand is an address or not. 
This provides us with useful information that is unavailable otherwise.

The functions provided to access the relocation table are:
\begin{description}
\item [IsAddressRelocatable: ADDRESS $\rightarrow$ BOOL].
	Given a native address, returns whether the address is in the
	relocation table or not.

\item [GetRelocatedAddress: ADDRESS $\rightarrow$ ADDRESS].
	Given a native relocatable address, returns the relocated 
	address for it. This function is tentative and not implemented
	as yet; RISC machines require many bit manipulations, and
	this will probably require a different interface function.
\end{description}

With object files, relocations are often associated with symbols.
Sometimes, (e.g. with GCC), intramodule function calls that
could be resolved at compile time are left until link time, so
the call has a relocation entry with an associated symbol table
entry (often not the same symbol table as the main symbols).
To make use of this, the BinaryFile provides this function:

\begin{description}
\item [GetRelocSym: ADDRESS $\rightarrow$ STRING]. Given a
native address, returns a symbol associated with the relocation
at that address, if any. If there is no symbol, the function
returns 0.
\end{description}


\subsection{Program Headers}
Application writers normally like dumping information about 
the headers of the program in order to determine format-specific
information that is not otherwise accessible via the loader
interface.  For example, a loader may have a verbose option to
display this type of information.

Unfortunately, it is difficult to provide a clean interface for
such low level details. (The previous version of the loader
attempted this, but it was not satisfactory). Therefore this
version of BinaryFile provides just one function, for dumping
all the headers in a verbose manner:

\begin{description}
\item [DisplayDetails: BinaryFile x STRING x FILE* $\rightarrow$ BOOL].
	Given a BinaryFile class and the name of the file, sends a verbose
	listing of the
	file to the file indicated by the FILE* parameter.
	The STRING parameter is merely to provide the first line of output,
	which is typically {\it Type} header for {\it Filename}.
	The FILE* parameter can be a special stream such as
	\texttt{stdio}, or a file opened with the \texttt{fopen}
	library function.

\end{description} 

The addresses referred to above are host addresses, i.e. pointers
to actual loaded data. (By contrast, native addresses are those
where part of a file is expected to be loaded in the native
operating system. Headers have no native address.

In the ElfBinaryFile, more specific functions are provided to
access different named headers, such as GetProgramHeader()
and GetSectionHeader().

Given the address of a header, an application writer would have
to know what the contents of the header looks like, and would
have to write the dump function.  We do not consider this
type of detail should be included in our BinaryFile class.


\subsection{Analysis Functions}
The following functions are provided for extra functionality 
required under binary translation, they are:
\begin{description}
\item [GetInitialState: BinaryFile $\rightarrow$ LIST$<$(REGxADDRESS)$>$].
	Returns a list of registers and their initial value (normally
	an address) prior to program execution.

\item [GetEntryPoints: BinaryFile $\rightarrow$ LIST$<$ADDRESS$>$].
	Returns a list of native addresses which are entry points into the 
	program.  The first address is always the initial value of the program
	counter (PC).

\emph{The following function is not implemented at present. It occured to
	me (Mike) that
	the mapping from Native to Host address may vary for each section;
	it doesn't in elf, but it may for other sections. Let's leave this
	one out unless we really need it. It is easy to implement anyway:
	just use the difference between native and host addresses in the
	section info.}
\item [NativeToHostAddress: (BinaryFile x ADDRESS) $\rightarrow$ ADDRESS].
	Given a native address, returns its equivalent host address.
	This function allows access to a block of bytes, since a native
	address is just a pointer.

\item [IsDynamicLinkedProc: (BinaryFile x ADDRESS) $\rightarrow$ BOOL].
	Given a native address, returns whether the address is that of a 
	dynamically linked-in procedure or not in this BinaryFile class.

\item [LoadDynamicLinkedProc: BinaryFile $\rightarrow$ ADDRESS $\rightarrow$ BOOL].
	Given the address of a dynamically linked-in procedure (i.e. the
	target of a call instruction that calls such a function, which
	will usually be an address in the Program Linkage Table), loads
	the appropriate source library file (if needed) to resolve this
	call. It returns its success or otherwise, and the native address
	where the procedure is loaded. 

	There will have to be environment variables to specify what path(s)
	should be searched to locate the appropriate library. These would
	be called \\
	X86\_LIBRARY\_PATH\\
	SPARC\_LIBRARY\_PATH\\
	and so on.  For example,
	if an X86 source program calls the \texttt{atan} function,
	the program's dynamic section specifies the DT\_NEEDED libraries
	as "libc.so.1" and "libm.so.1", and the environment variable
	X86\_LIBRARY\_PATH is set to "/var/lib/x86:/usr/lib/x86", then
	the following libraries will be searched for the \texttt{atan}
	function: \\
	/var/lib/x86/libc.so.1, /var/lib/x86/libm.so.1,
	/usr/lib/x86/libc.so.1, /usr/lib/x86/libc.so.1 \\
	Note that this function is only needed if/when we start supporting
	translations of libraries; at present we assume a native library
	is available.  Hence this function is not implemented at present.
\end{description}



\section{Notes on Individual BinaryFiles}

At present, the ElfBinaryFile is by far the most complete. It is the
only version that implements DisplayDetails(). The need to use
ElfDetails.cc at all, and all the verbose dumping code, can be 
avoided by defining \texttt{NODETAILS} in the make.

The ExeBinaryFile class does not handle symbols (e.g. Microsoft
Codeview or Borland) as these are stored in different ways by
different EXE compilers.

ElfBinaryFile implements symbols, both by address and by name. The
table of symbols looked up by value comes from the \texttt{.symtab}
section, if present, otherwise from the \texttt{.dynsym} section.

When looking up symbols by name, the \texttt{.hash} section is used. This
section refers to the \texttt{.dynsym} section, not the \texttt{.symtab}
section (if present). 
Thus, it is possible to look up local (non dynamic) symbols by value, 
but not by name.  This is a limitation of the way that elf files
are made. 
ElfBinaryFile implements the symbol size field of the SYM\_VALUE passed to 
ValueByName().

Some library files (e.g. libstdc++) have silly dynamic symbol table
entries (that they don't define), with a symbol type of STT\_NOTYPE.
Symbols of this kind are treated as if they don't exist in the symbol
table (i.e. ValueByName() returns 0 for these). 



\section{Interface Functions to Construct and use an ArchiveFile}
\label{sec-archive}

The ArchiveFile class and each of its parts (described in the abstract 
format description of a binary file in section~\ref{sec-bff}), are 
explained here in terms of interface functions provided for accessing 
information in those parts. 

\begin{description}
\item [ArchiveFile: nil $\rightarrow$ ArchiveFile].
	This is the constructor; there are no arguments. The object is
	created without a file being loaded; use the Load function below
	to load an archive file.

\item [Load: STRING $\rightarrow$ BOOL].
	Loads the archive file whose name is given. Returns false if the
	file could not be found, not opened, or it was not an archive
	file. Messages may be displayed on standard error.

\item [UnLoad: ArchiveFile $\rightarrow$ nil].
	Unloads the archive file, and any members that may have been loaded
	and not yet unloaded.  All information about the archive file
	and member files are invalid after calling this function.

\item [GetNumMembers: ArchiveFile $\rightarrow$ int].
	Returns the number of members, not including any special members
	that may exist merely for administrative purposes (e.g. the
	empty members in elf archives that hold the symbol table).

\item [GetMember: ArchiveFile x int $\rightarrow$ BinaryFile*].
	Given an index (0 for first, etc), constructs, loads, and returns
	a pointer to a BinaryFile object for that member. The binaryfile
	object comes pre-loaded; the user should not call the Load function
	for the returned member, but should call UnLoad when done with the
	member object. In the case of errors, a NULL pointer is returned.

\item [GetMemberByProcName: ArchiveFile x STRING $\rightarrow$ BinaryFile*].
	Similar to the above function, except that the relevant member is
	selected by procedure name, e.g. "arctan".

\item [GetMemberByFileName: ArchiveFile x STRING $\rightarrow$ BinaryFile*].
	Similar to the above function, except that the relevant member is
	selected by the name of the archive member (this will usually be an
	object file without an explicit path, e.g. "transcend.o").

\item [GetMemberFileName: ArchiveFile x int $\rightarrow$ STRING].
	Given a member index, return the name of the archive member.


\end{description}

Many of the functions above presume some knowledge of the contents of
the archive, e.g. the file name of the members, or symbols in those
members. There are at present no functions to allow the user to browse
these items.

\section{Example Code}

Examples are often worth a thousand words.  Here are two examples
on how to use the BinaryFile and derived classes.


\subsection{Example 1}
Here is a complete, though simple, program to use the ElfBinaryFile class. 
It simply iterates through the sections of the file, and counts how 
many sections contain code (i.e.  have the ST\_CODE flag set).

\begin{verbatim}
#include <stdio.h>
#include "ElfBinaryFile.h"

int main(int argc, char* argv[])
{
    ElfBinaryFile L;
    int iCnt = 0;

    if (!L.Load(argv[1])) return 1;
    int n = L.GetNumSections();
    PSECTIONINFO pSect = L.GetSectionInfo();
    for (int i=0; i < n; i++)
        if (pSect[i].bCode) iCnt++;
    printf("Part %d has %d code section(s)\n", iPart, iCnt);
    L.UnLoad();
    return 0;
}
\end{verbatim}

This can be compiled as follows (assuming that the above source is in
\texttt{sample.cc}):
\begin{verbatim}
gcc -o sample sample.cc BinaryFile.cc ElfBinaryFile.cc ElfDetails.cc
    SymTab.cc -lelf -lstdc++
\end{verbatim}

The \texttt{"-lelf"} is because ElfBinaryFile requires the elf library; other
versions of the loader would probably not need any libraries other than the
standard template library.
The results are:

\begin{verbatim}
% sample sample
Part 1 has 4 code section(s)

% sample /usr/lib/libelf.a
Part 1 has 1 code section(s)
Part 2 has 1 code section(s)
...
Part 42 has 1 code section(s)
\end{verbatim}


\subsection{Example 2}
The following example dumps the \texttt{\_exit} function
(if present) of the given input file.   Assume the following
code is in file dumptext.cc.

\begin{verbatim}
#include "ElfBinaryFile.h"

int main(int argc, char* argv[])
{
    ElfBinaryFile L;
    SYM_VALUE v;
    PSECTIONINFO pText;

    if (!L.Load(argv[1])) return 1;

    if (L.ValueByName("_exit", &v))
    {
        pText = L.GetSectionInfoByName(".text");
        if (pText)
        {
            int n = v.iSymSize;
            printf("Function _exit (%d bytes):\n", n);
            printf("%08X: ", pText->uNativeAddr);    // Address
            // Compute offset from start of text section to this function
            dword offset = v.uSymAddr - pText->uNativeAddr;
            dword p = pText->uHostAddr + offset;    // Start of function
            for (int i=0; i < n; i++)
                printf("%02X", *((unsigned char*)p++));
            printf("\n");
        }
    }
    else printf("No function '_exit'\n");
    return 0;
}
\end{verbatim}

To compile it, at the command line type:
\begin{verbatim}
gcc -g -o dumpexit dumpexit.cc BinaryFile.cc ElfBinaryFile.cc SymTab.cc 
    ElfDetails.cc -lelf -lstdc++
\end{verbatim}

Sample output: 
\begin{verbatim}
% dumpexit /usr/lib/libc.so.1 
Function _exit (8 bytes):
000169A0: 8210200191D02008
% 
\end{verbatim}

\subsection{Example 3}

The following example loads the archive "/usr/lib/libm.a" and displays
the load address for the "sin" function.

\begin{verbatim}
#include <stdio.h> 
#include "ElfArchiveFile.h"
void main()
{

    ElfArchiveFile af;
    if (!af.Load("/usr/lib/libm.a"))
    {
        printf("ArchiveFile could not load file\n");
        return 1;
    }

    printf("There are %d members in this archive\n", af.GetNumMembers());
    printf("First 5 file names:\n");
    for (int i=0; i < 5; i++)
    {
        printf("%s\n", af.GetMemberFileName(i));
    }

    BinaryFile* pm;
    printf("Object for function 'sin' is at %p\n",
        pm = af.GetMemberByProcName("sin"));
    printf("Address of sin is %p; size %d\n", pm->GetAddressByName("sin"),
        pm->GetSizeByName("sin"));
    pm->UnLoad();

	af.UnLoad();
}
\end{verbatim}

To compile, (assuming the example code is called example3.cc):
\begin{verbatim}
gcc -g -DUNIX -o example3 example3.cc BinaryFile.cc ElfBinaryFile.cc
    SymTab.cc ElfDetails.cc ArchiveFile.cc ElfArchiveFile.cc -lelf -lstdc++
\end{verbatim}


\subsection{Compiling and Linking}
Applications using the BinaryFile class need to compile the files 
\texttt{BinaryFile.cc} and \texttt{ExeBinaryFile.cc} (or \texttt{ElfBinaryFile.cc},
\texttt{PEBinaryFile.cc}, etc).  They should link the resultant object files 
with the application. The final executable must be linked with the standard
template library, e.g. "-lstdc++" for the GCC compiler. Elf versions
also require the elf library, i.e. "-lelf" for GCC.

In addition, BinaryFile classes implementing symbol tables (at present,
this is only the ElfBinaryFile class) also need to compile \texttt{SymTab.cc}
which depends on \texttt{SymTab.h}, and should link with \texttt{SymTab.o}.

For the ElfBinaryFile class, unless \texttt{NODETAILS} is defined in
the make (with \texttt{-DNODETAILS}), the file \texttt{ElfDetails.cc}
must also be compiled and the resultant \texttt{ElfDetails.o} linked.

In those application source files using the BinaryFile class, the appropriate
header file (but not \texttt{BinaryFile.h} itself) should be included. For
example, if using ElfBinaryFile:
\begin{verbatim}
#include "ElfBinaryFile.h"		// Definitions for the ElfBinaryFile class
\end{verbatim}

An example makefile fragment:
\begin{verbatim}
OBJS = myfile1.o ... BinaryFile.o ElfBinaryFile.o SymTab.o ElfDetails.o

ElfBinaryFile.o: BinaryFile.o ElfBinaryFile.h ElfDetails.h

SymTab.o: SymTab.cc SymTab.h

BinaryFile.o: BinaryFile.h
\end{verbatim}



